

Citation: Miao Z, Alvarez M, Ko A, Bhagat Y, Rahmani E, Jew B, et al. (2020) The causal effect of obesity on prediabetes and insulin resistance reveals the important role of adipose tissue in insulin resistance. PLoS Genet 16(9): e1009018. https://doi.org/10.1371/journal.pgen.1009018

Editor: Elizabeth R. Hauser, Duke University, UNITED STATES

Received: November 12, 2019

Accepted: July 29, 2020

Published: September 14, 2020

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pgen.1009018

Copyright: © 2020 Miao et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: This research has been conducted using the UK Biobank Resource under Application Number 33934. The UK Biobank data is available from the UK Biobank data RESEARCH ARTICLE

The causal effect of obesity on prediabetes and insulin resistance reveals the important role of adipose tissue in insulin resistance

Zong Miao^{1,2}, Marcus Alvarez¹, Arthur Ko³, Yash Bhagat¹, Elior Rahmani⁴, Brandon Jew^{2,4}, Sini Heinonen⁵, Linda Liliana Muñoz-Hernandez^{6,7,8}, Miguel Herrera-Hernandez⁹, Carlos Aguilar-Salinas^{6,7,8}, Teresa Tusie-Luna¹⁰, Karen L. Mohlke¹¹, Markku Laakso¹², Kirsi H. Pietiläinen^{5,13}, Eran Halperin^{1,4,14,15,16}, Päivi Pajukanta^{1,2,16}*

1 Department of Human Genetics, David Geffen School of Medicine at UCLA, Los Angeles, California, United States of America, 2 Bioinformatics Interdepartmental Program, UCLA, Los Angeles, California, United States of America, 3 Department of Medicine, David Geffen School of Medicine at UCLA, Los Angeles, California, United States of America, 4 Computer Science Department in the School of Engineering, UCLA, Los Angeles, California, United States of America, 5 Obesity Research Unit, Research Program for Clinical and Molecular Metabolism, Faculty of Medicine, University of Helsinki, Helsinki, Finland, 6 Unidad de Investigación en Enfermedades Metabólicas, Dirección de Nutrición, Instituto Nacional de Ciencias Médicas y Nutrición Salvador Zubirán, Mexico City, Mexico, 7 Departamento de Endocrinología y Metabolismo, Instituto Nacional de Ciencias Médicas y Nutrición Salvador Zubirán. Mexico City, Mexico, 8 Tecnologico de Monterrey, Escuela de Medicina y Ciencias de la Salud, Monterrey, Nuevo Leon, México, 9 Departamento de Cirugía, Instituto Nacional de Ciencias Médicas y Nutrición, Mexico City, Mexico, 10 Unidad de Biología Molecular y Medicina Genómica Instituto de Investigaciones Biomédicas UNAM / Instituto Nacional de Ciencias Médicas y Nutrición Salvador Zubiran, Mexico City, Mexico, 11 Department of Genetics, University of North Carolina, Chapel Hill, North Carolina, United States of America, 12 Institute of Clinical Medicine, Internal Medicine, University of Eastern Finland and Kuopio University Hospital, Kuopio, Finland, 13 Obesity Center, Endocrinology, Abdominal Center, Helsinki University Central Hospital and University of Helsinki, Helsinki, Finland, 14 Department of Computational Medicine, UCLA, Los Angeles, California, United States of America, 15 Department of Anesthesiology and Perioperative Medicine, UCLA, Los Angeles, California, United States of America, 16 Institute for Precision Health, David Geffen School of Medicine at UCLA, Los Angeles, California, United States of America

* ppajukanta@mednet.ucla.edu

Abstract

Reverse causality has made it difficult to establish the causal directions between obesity and prediabetes and obesity and insulin resistance. To disentangle whether obesity causally drives prediabetes and insulin resistance already in non-diabetic individuals, we utilized the UK Biobank and METSIM cohort to perform a Mendelian randomization (MR) analyses in the non-diabetic individuals. Our results suggest that both prediabetes and systemic insulin resistance are caused by obesity ($p = 1.2 \times 10^{-3}$ and $p = 3.1 \times 10^{-24}$). As obesity reflects the amount of body fat, we next studied how adipose tissue affects insulin resistance. We performed both bulk RNA-sequencing and single nucleus RNA sequencing on frozen human subcutaneous adipose biopsies to assess adipose cell-type heterogeneity and mitochondrial (MT) gene expression in insulin resistance. We discovered that the adipose MT gene expression and body fat percent are both independently associated with insulin resistance ($p \le 0.05$ for each) when adjusting for the decomposed adipose cell-type proportions. Next, we showed that these 3 factors, adipose MT gene expression, body fat percent, and adipose repository, but restrictions apply to the availability of these data, which were used under license for the current study and therefore are not publicly available. Data are however available with the permission of UK Biobank. The GTEx dataset analyzed during this study are available in the dbGAP repository, phs000424.v6.p1 (currently available under accession number phs000424.v8. p2). The METSIM gene expression data were previously made available in GEO under GSE135134.

Funding: This study was funded by National Institutes of Health (NIH) grants HL-095056, HL-28481, R01HG010505, and U01 DK105561. Z.M was supported by the AHA grant 19PRE34430112, M.A. was supported by the HHMI Gilliam grant. and A.K. by the NIH grant F31HL127921. E.H., E.R. and B.J. were partially supported by the National Science Foundation grant 1705197. E.H., E.R., and B.J. were partially supported by NIH/NHGRI HG010505-02. K.H.P. was supported by the Academy of Finland (272376, 266286, 314383, 315035), Finnish Medical Foundation, Finnish Diabetes Research Foundation, Novo Nordisk Foundation, Gyllenberg Foundation, Sigrid Juselius Foundation, Helsinki University Hospital Research Funds, Government Research Funds and University of Helsinki. The funders had no role in study design, data collection, and analysis, decision to publish, or preparation of the article.

Competing interests: The authors have declared that no competing interests exist.

cell types, explain a substantial amount (44.39%) of variance in insulin resistance and can be used to predict it ($p \le 2.64 \times 10^{-5}$ in 3 independent human cohorts). In summary, we demonstrated that obesity is a strong determinant of both prediabetes and insulin resistance, and discovered that individuals' adipose cell-type composition, adipose MT gene expression, and body fat percent predict their insulin resistance, emphasizing the critical role of adipose tissue in systemic insulin resistance.

Author summary

Obesity is a global health epidemic predisposing to type 2 diabetes (T2D) and other cardiometabolic disorders. Previous studies have shown that obesity has a causal effect on T2D; however, it remains unknown whether obesity causes prediabetes and insulin resistance already in non-diabetic individuals. By utilizing almost half a million individuals from the UK Biobank and the Finnish METSIM cohort, we identified a significant causal effect of obesity on prediabetes and insulin resistance among the non-diabetic individuals. Next, we investigated the role of subcutaneous adipose tissue in these obesogenic effects. We discovered that the adipose mitochondrial gene expression and body fat percent are independently associated with insulin resistance after adjusting for the tissue heterogeneity. For the latter, we estimated the adipose cell type proportions by utilizing single-nucleus RNA sequencing of frozen adipose tissue biopsies. Moreover, we established a prediction model to estimate insulin resistance using body fat percent and adipose RNA-sequencing data, which enlightens the importance of adipose tissue in insulin resistance and provides a helpful tool to impute the insulin resistance for existing adipose RNA-sequencing cohorts. Overall, we discover the potential causal effect of obesity on prediabetes and insulin resistance and the key role of adipose tissue in insulin resistance.

Introduction

The global obesity epidemic is driving the concomitant rapid increase in the prevalence of cardiometabolic disorders, including type 2 diabetes (T2D) [1,2]. It is well established that obesity, prediabetes, and insulin resistance are tightly associated [3-8]. Although prediabetes is an inevitable stage for T2D patients, some people with prediabetes do not ultimately develop diabetes given that the annualized conversion rate of prediabetes to diabetes is estimated to be 5%–10% [9]. Thus, it is important to investigate whether obesity is also a causal risk factor for the early stages of T2D development (i.e. prediabetes). Moreover, inflammation has been identified as the link between obesity and insulin resistance [10-15]. For example, Roberts-Toler et al. showed that diet-induced obesity can cause insulin resistance in mouse brown adipose tissue [16]. Wensveen et al. showed that natural killer cells can mediate the association between obesity and insulin resistance in mice [17]. However, the direction of the causal effect between obesity and insulin resistance remains elusive in humans [18,19]. Thus, direct evidence of obesity causing systemic insulin resistance in humans is still lacking. To this end, we performed a Mendelian randomization (MR) analysis using the genotype and metabolic traits of unrelated non-diabetic individuals from both UK Biobank (UKB) [20] and the Finnish METabolic Syndrome In Men (METSIM) cohort [21]. This report suggests for the first time using MR that obesity (i.e. body fat percent) is causally related with both prediabetes and insulin resistance in non-diabetic humans.

The key functions of adipose tissue, i.e. lipogenesis (storing fat) and lipolysis (mobilizing the stored fat), make it one of the most important tissues contributing to obesity. Thus, it would be important to better understand how much this endocrine tissue contributes to insulin resistance and T2D. Since adipose tissue is complex and contains multiple cell-types, adipose cell-type composition may be affected by obesity. Weisberg et al. showed that the number of macrophages increases in the adipose tissue of obese mice [22]. Furthermore, in human adipose tissue, high BMI was reported to be negatively correlated with the number of adipocytes [23] and positively correlated with the size of the adipocytes [22–24]. However, the effects of different adipose cell-type proportions on insulin resistance have not been systematically assessed in humans previously.

Fluorescence-activated cell sorting (FACS) has been used for characterizing and defining some of the cell types in human adipose samples [25–27]. Even though the number of marker proteins that can be simultaneously measured in FACS has progressively increased [28], FACS relies on predetermined cell-type specific marker proteins to isolate different cell types and is thus unable to discover new cell types or sub-cell types. Moreover, since FACS requires a large starting number of cells (more than 10,000) in suspension [29], it is unable to isolate single cells from a low quantity cell population. Overall, it is highly challenging to evaluate all cell types of solid tissues, such as adipose tissue, using FACS. Thus, to thoroughly investigate the tissue heterogeneity in human adipose tissue, we performed single nucleus RNA sequencing (sn-RNA-seq) of all adipose cell types using frozen human subcutaneous adipose biopsies. We then utilized the sn-RNA-seq data to define expression profiles of signature genes in different adipose cell-type to decompose cell-type proportions in the bulk adipose RNA-seq cohorts. This helped us leverage the gene expression information available in the adipose bulk RNA-seq data to assess whether adipose cell-type composition influences systemic insulin resistance.

Previous studies have shown that the biogenesis and metabolic activities of the mitochondria (MT) are impaired in the adipose tissue of obese individuals [2, 30–33]. Experimental evidence also shows that declined MT function can elicit insulin resistance in mice [34,35]. Paglialunga et al. further demonstrated that elevated MT reactive oxygen species (ROS) emission in murine white adipose tissue contributes to insulin resistance [35]. Furthermore, dysfunction of MT in muscle and liver associates with insulin resistance in humans [36,37]. However, since the MT activity, obesity status (body fat percent or BMI), and systemic insulin resistance are associated with each other, it is unclear whether these associations are caused by independent mechanisms or confounded by a shared trait. To this end, we investigated whether MT gene expression in human adipose tissue is independently associated with systemic insulin resistance and body fat percent/BMI. Furthermore, we built a prediction model to investigate whether systemic insulin resistance (i.e. assessed using the Matsuda index) in humans can be predicted using adipose cell-type proportions, adipose MT gene expression, and body fat percent as an input. Overall our studies helped determine the causal role of obesity in human insulin resistance, of which a major portion is driven by adipose tissue.

Results

Evidence from Mendelian randomization for the role of obesity in prediabetes and systemic insulin resistance in non-diabetic individuals

Although obesity and prediabetes are known to be associated [7,8], there is no previous MR evidence about the causal direction between them in non-diabetic individuals. To this end, we performed an MR analysis to investigate whether prediabetes (assessed by serum HbA1C level between 5.7–6.4 [38]) is caused by obesity. Fig 1A shows the two-sample MR models we used to explore causal associations between prediabetes and obesity (body fat percent).



Fig 1. MR analysis shows the causal relationship of body fat percent on prediabetes and Matsuda index, i.e. obesity leads to insulin resistance. (A) Workflow of our MR analysis in UKB and METSIM cohorts. (B) The variant effect sizes on the exposure (i.e. body fat percent) are associated with the variant effect sizes on the outcome (i.e. the Matsuda index). The slope indicates the estimated causal effect of the exposure on the outcome. The label boxes showed the MR results after MR-PRESSO adjusted for potential pleiotropy. P.pleio indicates the p-value of pleiotropy identified by MR-egger. The IVs are significant body fat percent GWAS variants in UKB (see the <u>Methods</u> for details). (C) A two-sample MR analyses show the causal effect of obesity (i.e. body fat percent) on prediabetes in UKB cohort. The IVs are significant prediabetes GWAS variants in UKB (see the <u>Methods</u> for details). (D) A two-sample MR analysis did not identify the causal effect of prediabetes on obesity (i.e. body fat percent). The p-values of the causal effects from MR-PRESSO and MR-egger were adjusted for 3 tests using a Bonferroni correction. To keep a stringent control of the potential pleiotropy, the other p-values (Heterogeneity test and P.pleio) were not adjusted for multiple testing. NS indicates a non-significant p-value (p-value > 0.05). The IVs are significant body fat percent GWAS variants in UKB (see the <u>Methods</u> for details).

https://doi.org/10.1371/journal.pgen.1009018.g001

As BMI reflects both fat and lean mass, we used body fat percent instead of BMI in our MR analyses to utilize a more specific surrogate of obesity [39,40]. For this MR analysis, we first randomly separated the UKB cohort into two independent groups (n = ~190k in each group). Then we performed a GWAS for body fat percent in one of the UKB groups that identified 74 non-redundant SNPs ($R^2 < 0.01$) significantly associated with body fat percent

 $(p < 5 \times 10^{-8})$, which we used as the genetic instrumental variables (IVs) in the MR analysis (see Methods). Next, we used the second UKB group to perform a GWAS analysis on prediabetes to fulfill the two-sample MR design. Since dyslipidemia and abnormal blood pressure are known factors associated with obesity and diabetes [41,42], we excluded the IVs that are significantly associated with diastolic/systolic blood pressure or serum total triglycerides in the UKB cohort to avoid potential pleiotropy. Finally, we used the summary statistics from the GWAS analysis and MR-PRESSO [43] to identify a significant positive causal effect of body fat percent on prediabetes in the UKB (estimate effect = 0.039; adjusted pvalue = 1.2×10^{-3}). To further investigate the effect of obesity on prediabetes, we also employed MR-egger and heterogeneity test on the IVs adjusted by MR-PRESSO. Although MR-egger did not identify a significant causal effect of body fat percent on prediabetes using the threshold of p-value < 0.05, neither MR-egger nor heterogeneity test identified a significant pleiotropy in this MR analyses (Fig 1C). Since MR-egger usually tend to have less power compared to other MR methods [44], finding no evidence for pleiotropy from MR-egger and heterogeneity test appears still to be in line with the causal effect of body fat percent on prediabetes identified by MR-PRESSO (Fig 1C). This also suggests that our MR model fulfills the second and third assumptions of MR analysis (i.e. the IV is not associated with the hidden confounders or with the outcome variable (i.e. prediabetes) when conditioning on the exposure variable (i.e. body fat percent)). To test the possibility of the reverse causal path, we utilized 47 independent prediabetes GWAS SNPs as IVs from our GWAS in the second half of the UKB and explored the potential causal effect of prediabetes on obesity. Fig 1D shows that there is no causal effect of prediabetes on body fat percent (estimated effect = 0.13, adjusted p-value > 0.05). As we performed 3 MR analyses to explore the causal effects among obesity, insulin resistance and prediabetes, we adjusted the p-values of causal effects for 3 tests using a Bonferroni correction. Thus, using MR in the UKB, we established a one-way causal effect of obesity on prediabetes.

To further investigate this finding, we next explored the causal relationship between obesity and systemic insulin resistance (i.e. decreased insulin sensitivity assessed by the Matsuda index) in the METSIM cohort. We used the 241 significant non-redundant body fat percent GWAS SNPs identified in the UKB as IVs for the MR analyses in METSIM. Using MR-PRESSO, we discovered a negative causal effect of body fat percent on the Matsuda index (i.e. insulin sensitivity) in METSIM (estimate effect = -0.67, p-value = 3.1×10^{-24}). It is worth noting that MR-PRESSO did not find any evidence of global pleiotropy (p-value > 0.05) when testing the causal effect of body fat percent on the Matsuda index in METSIM (Fig 1B). Moreover, MR-egger verified this finding (adjusted p-value = 4.2×10^{-8}) while it identified no evidence of pleiotropy (P.pleio > 0.05). A heterogeneity test also showed the concordant results with MR-egger and MR-PRESSO (p-value > 0.05). When investigating the opposite direction of causality (i.e. insulin resistance -> obesity) using a similar pipeline, we found no genomewide significant SNPs associated with the Matsuda index in METSIM or other cohorts of previous studies. Furthermore, no insulin resistance parameters are measured in the UKB. Therefore, the first assumption of MR (i.e. IV is associated with the exposure variable) cannot be fulfilled. As METSIM may be underpowered to identify genome-wide significant SNPs for the Matsuda index, the current sample size of the Matsuda index GWAS does not allow a reliable MR analysis in this opposite direction. Accordingly, assessment of this direction using MR warrants further investigation in larger GWAS cohorts with the Matsuda index available for study.

To investigate whether potential confounders affect our MR analyses, we tested for association between the IVs and 3 potential confounders, i.e. alcohol intake, smoking, and physical activity. <u>S1 Fig</u> shows that in all 3 MR analyses, only 4 IV variants (rs7187776, rs1788783, rs1872841, and rs62037364) showed a significant association with these confounders (S1A Fig), and removing these 4 variants did not affect our MR results (S1B and S1C Fig), and thus obesity still showed a significant causal effect on both insulin resistance and prediabetes.

In summary, we established a one-way causal effect of obesity on prediabetes among the non-diabetic individuals from the UKB. Although MR-egger did not identify a similar significant causal effect of body fat percent on prediabetes as MR-PRESSO, the pleiotropy test employed by MR-egger and a heterogeneity test did not identify evidence for pleiotropy, which suggests no violation of the no-pleiotropy assumptions required by an MR analyses. In contrast, prediabetes did not show any evidence for a causal effect of body fat percent. We then followed up this finding by identifying a negative causal effect of body fat percent on insulin sensitivity (i.e. Matsuda index) in the non-diabetic individuals from METSIM. Although METSIM is underpowered to investigate the reverse-causal effect (i.e. insulin resistance -> body fat percent), the MR analyses performed in both UKB and METSIM suggest that obesity causes prediabetes and insulin resistance before the development of diabetes and thus, prediabetes is less likely to cause obesity among the non-diabetic population.

Adipose mitochondrial (MT) gene expression plays a key role in insulin resistance

We have shown that obesity (i.e. a high body fat percent) leads to increased insulin resistance in human using MR analysis. Since obesity reflects the amount of body fat, adipose tissue may play an important role in obesity-induced insulin resistance. Thus, we further investigated how adipose tissue affects the Matsuda index in the METSIM cohort. Among the 4k unrelated individuals in METSIM, 335 had bulk RNA-seq data from the subcutaneous adipose tissue biopsies. To estimate MT gene expression, we estimated the transcripts per million (TPM) values of each gene in the RNA-seq data and used the sum of TPMs from all 37 MT encoded genes to represent the MT gene expression. We also included the first 3 genetic PCs as covariates when correcting the MT expression to adjust for the potential population stratification (see Methods). We first observed that MT expression is significantly associated with BMI and body fat percent in the Finnish METSIM cohort using a threshold of p-value < 0.05 (S2A and S2B Fig). We also verified this association between the obesity stage and MT expression in another population, and our data in the Mexican cohort (see Methods) show that the lean Mexicans have significantly higher MT adipose expression when compared to the obese Mexicans (p-value = 0.016) (S2C Fig).

Next, we corrected the Matsuda index in METSIM for age, age^2 and excluded people who have type 2 diabetes (T2D) (n = 11). The Matsuda index and MT gene expression were inverse normal transformed to obtain normal distribution. Using these data, we discovered that the MT gene expression is significantly associated with Matsuda index (p-value = 9.60×10^{-15} , n = 324) in METSIM (Fig 2A). To further replicate and validate this finding, we tested the association between MT gene expression and insulin resistance in the RNA-seq data from 5 different tissues in the GTEx cohort. Since the GTEx cohort does not have the Matsuda index measured, we tested the MT gene expression difference between the patients with T2D and non-diabetic individuals. Fig 2B shows that, as in METSIM, the patients with T2D in GTEx have significantly lower MT gene expression in the subcutaneous and visceral adipose tissue than the non-diabetic individuals (S3A Fig). However, in three other non-adipose tissues from GTEx, only muscle MT gene expression (n = 305) showed the significant difference between T2D patients and non-diabetics (P = 3.42×10^{-2}). In the liver and whole blood (total n = 412), no significant difference was observed using the threshold of p-value < 0.05 (S3 Fig). These results support the important role of adipose tissue and muscle in the insulin resistance related



Fig 2. A low adipose MT gene expression is associated with insulin resistance. (A) In METSIM, the adjusted adipose MT gene expression is significantly associated with the adjusted Matsuda index. (B) In GTEx, the non-diabetic individuals have significantly higher adjusted MT gene expression (inverse normal transformed) than the T2D patients.

https://doi.org/10.1371/journal.pgen.1009018.g002



Fig 3. Analysis of sn-RNA-seq data reveals tissue heterogeneity in human subcutaneous adipose tissue. (A) We identified 8 cell-type clusters in 15,623 nuclei from frozen human adipose tissue from 6 Finnish individuals. The t-SNP plot is colored by the identified cell types. (B) Using sn-RNA-seq as reference, the estimated adipose cell-type proportions from bulk adipose RNA-seq data are well concordant with the true cell-type proportions.

https://doi.org/10.1371/journal.pgen.1009018.g003

metabolic process; however, due to the limited sample size of the GTEx liver cohort (n = 123), we cannot exclude the potential role of liver in insulin resistance. Taken together, the adipose MT gene expression is significantly associated with insulin resistance in the METSIM cohort and T2D in the GTEx cohort. In both subcutaneous and visceral adipose tissue, the MT gene expression is significantly lower in insulin resistant individuals.

Assessing tissue heterogeneity and adipose cell type proportions using single nucleus RNA sequencing

Single nucleus RNA sequencing reveals 8 cell-types in human adipose tissue. Adipose tissue is a complex tissue that consists of multiple cell-types, such as adipocytes, preadipocytes, macrophages, fibroblasts, and vascular cells. Even though adipocytes comprise ~90% of the total volume in the human adipose tissue, they only take ~50% of the total cell count [45–47]. We hypothesized that the metabolic processes in different contexts and adipose cell-types associated with obesity may substantially affect systemic insulin resistance. To investigate how adipose cell-type heterogeneity affects insulin resistance, we performed single-nucleus RNA sequencing (sn-RNA-seq) on 6 frozen human subcutaneous adipose tissue biopsies (see the Methods) and used cell-type in order to estimate cell-type proportions from the bulk adipose RNA-seq profiles.

Using the 10x Genomics platform, we sequenced on average ~2,600 nuclei for each sample and obtained the non-zero expression of ~500 genes per cell (for sample-specific metrics, see S1 Table). Next, we used Seurat to cluster the sn-RNA-seq data and identified 8 adipose cell-type clusters based on the gene expression profiles of the adipose nuclei. It is worth noting that the adipocyte cluster comprises 44.0% of the total cell number, which is in line with the previous findings [47]. Fig 3A shows the tSNE plots of the 8 adipose cell-type clusters in 15,623 nuclei. S4A Fig shows the tSNE plot that is colored by the sample IDs. These data show that clustering is largely driven by distinct gene expression profiles from different adipose cell-types rather than by the differences between individuals.

Estimating adipose cell-type proportions using the sn-RNA-seq as the reference. Next, we used MuSiC to estimate the proportions of each cell-type from the bulk adipose RNA-seq data. Utilizing both bulk RNA-seq and sn-RNA-seq data from these 6 individuals, we estimated the cell-type proportions of the 6 individuals from bulk RNA-seq data and compared the decomposition results with the true cell-type proportions from the sn-RNA-seq data to verify our decomposition method. We employed a leave-one-out approach to decompose the cell-type proportions of each sample while using the sn-RNA-seq data of the other 5 samples as the reference. Fig 3B shows that the estimated adipose cell-type proportions have a high concordance with the true adipose cell-type proportions. Thus, our decomposition method provides reliable estimated adipose cell-type proportions then we used the 6 sn-RNA-seq samples as reference. S2 Table lists the true cell-type proportions of the 6 sn-RNA-seq samples.

After verifying the accuracy of cell-type decomposition by MuSiC, we applied the method to the 335 subcutaneous adipose bulk RNA-seq samples from the METSIM cohort to estimate the proportions of the 8 adipose cell-types (S3 Table). We first checked the associations between the Matsuda index and the 8 estimated cell-type proportions using linear regression. In the association tests, we inverse normal transformed the Matsuda index and included age as a covariant. Four of 8 estimated cell-type proportions showed a significant association (Bonferroni adjusted p-value < 0.05) with the Matsuda index (S4 Table). It is worth noting that both dendritic cells and macrophages exhibited a strong association with the Matsuda index (p-

values $< 5 \times 10^{-5}$), suggesting that immune cell types contribute to insulin resistance in human adipose tissue.

Body fat percent and adipose MT gene expression are independently associated with systemic insulin resistance after adjusting for tissue heterogeneity. It has been shown previously that a high BMI associates with adipose MT activity [48]. In line with this we observed that MT gene expression is significantly associated with body fat percent (p-value = 3.55×10^{-7}) in the adipose RNA-seq data from METSIM (n = 324). However, it is unknown if body fat percent or adipose tissue heterogeneity causes the association between the adipose MT expression and insulin resistance. To investigate this, we explored the associations between the Matsuda index and age, body fat percent, adipose MT gene expression, and adipose cell-type proportions using a multi-variable linear model (see model 1 in Methods). In our multi-variable linear model, body fat percent, MT expression, and the estimated proportions of dendritic and fibroblasts cells in adipose all showed significant associations with the Matsuda index (p < 0.05) (S5 Table). This result suggests that the adipose tissue heterogeneity, MT gene expression, and body fat percent all have independent contributions to the variance in the Matsuda index. Noteworthy, this model explained 43.37% of the variance (R^2) in the Matsuda index, which is higher than using any trait alone: body fat percent ($R^2 = 30.89\%$); MT expression ($R^2 =$ 14.64%); and estimated cell-type proportions ($R^2 = 29.24\%$). Moreover, when we excluded body fat percent from model 1, the estimated adipose cell-type proportions and MT expression together explained a substantial amount ($R^2 = 35.42\%$) of the variance in the Matsuda index. The high variance explained by model 1 makes it possible to predict the Matsuda index, i.e. systemic insulin resistance, using body fat percent, adipose MT gene expression and the estimated adipose cell-type proportions.

Utilizing adipose RNA-seq data to predict systemic insulin resistance

Although the Matsuda index is an important biomarker for glucose metabolism, its measurement requires an oral glucose tolerance test, which is not available in many adipose RNA-seq cohorts. To this end, we developed a prediction model using elastic net regularization [49] that combines body fat percent/BMI, MT gene expression, age, and cell-type proportion information to predict the Matsuda index using adipose RNA-seq data. Although body fat percent is a more ideal trait for obesity compared to BMI, BMI is easy to measure, which makes it widely available in most human adipose RNA-seq cohorts. Thus, we trained two models that use either BMI or body fat percent to present obesity status to predict Matsuda index in METSIM. Employing a 100-fold cross validation in both prediction models, S5 Fig shows that the predicted Matsuda index has a high concordance with the true Matsuda index (r > = 0.64, pvalue $< = 1.65 \times 10^{-38}$) in both models. Since body fat percent and BMI exhibited a similar accuracy in predicting Matsuda index, we further tested the BMI model, which can potentially be applied to more cohorts. To further confirm this promising prediction of insulin resistance, we used the METSIM cohort to train a model and then estimated the Matsuda index in two independent adipose RNA-seq cohorts: GTEx and FTC (see Methods). Fig 4B shows that in the GTEx subcutaneous adipose samples, the T2D patients have significant lower predicted Matsuda index when compared to the non-diabetic GTEx individuals (p-value = 2.62×10^{-5}). Since the monozygotic twin participants share the identical genetic background, we tested the association between the predicted Matsuda index and the true Matsuda index both in the full FTC cohort and the unrelated individuals by randomly selecting one individual from each twin pair. Fig 4C and 4D shows that the predicted Matsuda index is similarly well concordant with the true Matsuda index in the full FTC cohort (r = 0.51, p-value = 1.15×10^{-7}) and in the unrelated FTC individuals (r = 0.46, p-value = 1.03×10^{-3}).



Fig 4. The predicted Matsuda index is well concordant with the true Matsuda index values in 3 different cohorts: METSIM, GTEx, and FTC. (A) In METSIM, the estimated and true Matsuda index are significantly associated. (B) In GTEx, the predicted Matsuda index is significantly higher in non-diabetic individuals when compared to the T2D patients. (C) In FTC, the estimated and true Matsuda index are significantly associated. (D) We randomly choose one individual from each twin pair to select the unrelated individuals from FTC. Among the unrelated individuals, the estimated and true Matsuda index are also significantly associated, indicating that the twin status did not bias the prediction of the Matsuda index.

https://doi.org/10.1371/journal.pgen.1009018.g004

Furthermore, the predicted Matsuda index had the best concordance with the true Matsuda index when compared to any of the tested traits alone in METSIM and FTC. <u>S6 Fig</u> shows that neither MT gene expression nor BMI can predict the Matsuda index as accurately as our prediction model in the METSIM and FTC cohorts. Therefore, this prediction model can

potentially be used to impute systemic insulin resistance into other adipose RNA-seq cohorts, in which this key glucose metabolism trait has not been measured. <u>S6 Table</u> shows the betas in the trained prediction model that can be applied to other cohorts. In summary, we discovered that a substantial amount (44.39%) of the variance in the systemic insulin resistance, measured using the Matsuda index, can be explained by adipose MT gene expression, adipose cell-type proportions, and BMI. By combining the information from these traits, we were repeatedly able to predict the Matsuda index with a great accuracy when compared to the prediction results with any single trait alone. Since the Matsuda index is an important biomarker for glucose metabolism in humans, our prediction model can be utilized to impute the Matsuda index index into adipose RNA-seq cohorts where this key metabolic trait is missing.

Discussion

Even though the previous MR studies have shown that obesity has a causal effect on T2D [50,51], MR evidence showing that obesity is causally linked with prediabetes or insulin resistance among non-diabetic individuals is missing. Noteworthy, the annualized conversion rate of prediabetes to diabetes is estimated to be 5%-10% [9] and thus, some prediabetes patients do not develop T2D. Thus, the causal effect between obesity and prediabetes cannot be simply deduced by the causal role of obesity on T2D. In the present study, we utilized the extensive UK Biobank cohort [20] and carefully phenotyped Finnish METSIM cohort [21] to investigate whether obesity causes prediabetes and causally increases insulin resistance in the non-diabetic population using the MR analysis. Our MR result sheds new light on the long-standing reverse causality question between obesity and insulin resistance by establishing its directionality. Stancakova et al. have showed earlier that the Matsuda index is the best index of insulin sensitivity when compared to other surrogate indexes of insulin resistance using an M value from the euglycemic hyperinsulinemic clamp as the gold standard [52]. Therefore, the Matsuda index is largely a measure of systemic rather than adipose-based insulin resistance. However, when we examined the role of adipose cell-type heterogeneity, adipose MT gene expression, and BMI in systemic insulin resistance, we discovered that even when excluding BMI from the calculation, the estimated adipose cell-type proportions and adipose MT gene expression together still explain a substantial amount ($R^2 = 35.42\%$) of the variance in the Matsuda index. When we included BMI into this analysis, all three factors are independently associated with insulin resistance (p < 0.05 for each) and the R² increased to 44.39% which is higher than using any trait alone ($R^2 < = 30.89\%$). This surprisingly high proportion of variance explained by adipose tissue (i.e. adipose cell types and MT gene expression) and BMI suggests that adipose tissue has an important role in the systemic insulin resistance. Based on this novel finding, we built a prediction model using adipose cell-types, adipose MT gene expression, and BMI that accurately predicted insulin resistance across multiple cohorts.

To investigate how adipose tissue heterogeneity affects systemic insulin resistance, we performed sn-RNA-seq using 6 frozen human subcutaneous adipose tissue samples. Noteworthy, the previous studies investigating adipose cell-type heterogeneity used FACS [25–27]; however this application is limited to a small number of well identified non-adipocyte cell-types and is unable to detect refined new subtypes. There are no previous publications performing sn-RNA-seq from frozen human adipose tissue, and thus the role of cell-type heterogeneity in insulin resistance has not been investigated for all main adipose cell types before. After careful quality control, the sn-RNA-seq generated 15,623 nuclei from the 6 adipose tissue biopsies and identified 8 adipose cell-type clusters based on their gene expression. We then used the sn-RNA-seq data as the reference data to detect cell-type specific signature genes in each adipose cell type cluster and decomposed the cell-type proportions in the METSIM, FTC, and GTEx adipose bulk RNA-seq cohorts, leveraging thus substantially the information contained in these existing bulk RNA-seq cohorts. Notably, the estimated cell-type proportions of macrophages and dendritic cells exhibited a significant association with insulin resistance, demonstrating the key role of the obesity-induced low-grade inflammation process in systemic insulin resistance [10,11,13–15].

Even though insulin resistance is an essential clinical metabolic trait in obesity-related cardiometabolic diseases, it is often not measured in the existing adipose RNA-seq cohorts, such as the GTEx cohort [55]. Moreover, although the adipose tissue is suggested to be relevant in the development of insulin resistance [12,16,34], to the best of our knowledge, the variance in insulin resistance parameters that can be explained by adipose tissue has not been reported previously. Strikingly, we found that 44.39% of the variance in systemic insulin resistance (i.e. the Matsuda index) can be explained by the adipose cell-types, adipose MT expression, and BMI using the METSIM cohort. Thus, we developed an elastic net prediction model to predict the Matsuda index using these traits. The prediction model was trained in a subset of the MET-SIM cohort. The model not only successfully predicted the Matsuda index in the METSIM test cohort but also predicted well the Matsuda index in 2 independent cohorts, the GTEx and FTC, indicating that we can predict the missing systemic insulin resistance estimates to cohorts lacking metabolic phenotype data, such as GTEx. Since the prediction model is based on adipose RNA-seq data, the predicted Matsuda index can potentially be used to proportionally estimate how much of insulin resistance is driven by adipose tissue versus other metabolic tissues. This would help subtype different forms of insulin resistance states underlying the development of type 2 diabetes.

As we used RNA-seq data to predict the cell-type proportions and then utilized those for the prediction of the Matsuda index, this two-step prediction design may introduce noise from both steps. However, utilizing gene expression across different cohorts is known to be prone to technical biases and batch effects [53], whereas the estimated cell-type proportions represent more uniform traits when compared to the gene expression. Moreover, as we have shown that the estimated cell-type proportions have relatively high associations with the true cell-type proportion, the cell-type proportions estimated by other methods, such as FACS sorting, can potentially also be used to predict Matsuda index in other studies.

In summary, we have shown that obesity has a strong participation in prediabetes and insulin resistance using the MR analysis. By leveraging bulk RNA-seq data in large adipose RNAseq cohorts using a small amount of adipose sn-RNA-seq data to decompose adipose celltypes, we show that a substantial proportion (44.39%) of systemic insulin resistance can be explained by certain adipose cell-type proportions, MT gene expression, and BMI. This new finding not only establishes the key role of adipose tissue in regulating insulin resistance but also provides a useful method to impute insulin resistance estimates to human transcriptome cohorts.

Materials and methods

Ethics statement

The METSIM, MOSS and FTC study designs were approved by local ethics committees and all participants gave a written informed consent.

Study cohorts

We analyzed the genotype and phenotype data of \sim 510k individuals from two cohorts: MET-SIM cohort (n = 10,198) [21], and UK Biobank cohort [20] (n = 391,816) In the METSIM cohort, middle-aged Finnish males were recruited at the University of Eastern Finland and

Kuopio University Hospital, Kuopio, Finland, and the biochemical lipid, glucose, and other clinical and metabolic phenotypes were measured as described previously [21]. Briefly, a 2-h oral glucose tolerance test (OGTT) (75 g of glucose) was performed in the METSIM cohort, and samples for plasma glucose and insulin were drawn at 0, 30, and 120 min [21]. We evaluated insulin resistance in the non-related, non-diabetic METSIM participants using the Matsuda index that was calculated based on the OGTT values, as described in detail previously [54]. The METSIM study design was approved by local ethics committee and all participants gave a written informed consent. This research has been conducted using the UK Biobank Resource under Application Number 33934. The UK Biobank data was downloaded from the UK Biobank data repository on 08/23/2018.

We analyzed the RNA-seq data in 751 human subcutaneous adipose samples from 4 different cohorts: METSIM cohort (n = 335) [21], Genotype-Tissue Expression (GTEx) cohort (n = 308) [55,56], the ongoing MOSS cohort (n = 107), and 54 monozygotic Finnish monozygotic twin cohort (FTC) (n = 108) [31,57,58]. The METSIM, MOSS and FTC study designs were approved by local ethics committees and all participants gave a written informed consent. The GTEx adipose RNA-seq data were downloaded from dbGaP (accession number phs000424.v6.p1) on 08/11/2016. In addition to subcutaneous adipose tissue, we also analyzed the GTEx visceral adipose tissue, blood, liver, and muscle RNA-seq data (v7). In the MOSS cohort, we collected subcutaneous adipose tissue biopsies from extreme obese Mexican individuals undergoing bariatric surgery in Instituto Nacional de Ciencias Medicas y Nutricion (INCMN), Mexico City. Currently, we have RNA-seq data from subcutaneous adipose samples from 88 obese Mexican MOSS participants and 19 lean Mexicans (BMI < 25) as controls. We used the FTC cohort to verify the prediction model of the Matsuda index. In this cohort, we generated RNA-seq data from subcutaneous adipose tissue of 54 MZ twin pairs (n = 108). S7 Table shows the clinical characteristics of the participants in the 4 cohorts. We also selected the adipose biopsies of 6 individuals from FTC for the sn-RNA-seq experiment. S1 Table shows the phenotypic characteristics of the 6 Finnish individuals whose adipose biopsies were processed for sn-RNA-seq. The 6 individuals have roughly similar ages, 3 of the 6 are males, and 3 of the 6 have a normal BMI (BMI<25).

GWAS and Mendelian randomization (MR) analysis

To identify candidate instrumental variables (IVs) for the two-sample MR analysis in UKB, we first randomly separated the UKB cohort into two independent groups and performed GWAS analyses of body fat percent and prediabetes in different groups. The prediabetes cases were identified by serum HbA1c level between 5.7-6.4 [38]. We excluded the individuals who had HbA1c > 6.4 or had been diagnosed as diabetic to ensure that only non-diabetic individuals were included in the GWAS analyses. We used BOLT-LMM [59] to explore the associations between the genotypes and the target phenotype, while accounting for the population stratification. We inverse normal transformed body fat percent to ensure a normal distribution and included age, age², sex, array type, center ID, and 20 genotype PCs as covariates. The percent of prediabetes cases in UKB is 14% (62,866 out of 441,057), which is in line with the BOLT-LMM recommendation of = > 10% case fraction for an unbiased case-control GWAS study [60]. To fulfill the first assumption of MR (i.e. IVs should be significantly associated with the exposure variable), we selected the independent (R2<0.01) GWAS SNPs (p-value<5e-8 in UKB) of body fat percent and prediabetes as candidate IVs for the MR analysis. Then we used MR-PRESSO [43] to identify causal associations between body fat percent and prediabetes while controlling for the potential pleiotropy. We also used MR-egger [61] and heterogeneity

test (Cochran's Q test) to verify that no pleiotropy biased our MR analysis. <u>S8 Table</u> shows the summary statistics employed in the MR analyses.

When searching for a causal effect of body fat percent on insulin resistance, we performed GWAS analyses of body fat percent and Matsuda index using the non-diabetic individuals METSIM using BOLT-LMM [59] following the same pipeline as in UKB. For these GWAS analyses, we inverse normal transformed body fat percent and Matsuda index and included age, age², and 10 genotype PCs as the covariates. In the GWAS, we did not identify any Matsuda index-associated SNPs in METSIM using a genome-wide significant cut point of pvalue $<5.0\times10^{-8}$. For body fat percent, we performed a GWAS analysis on body fat percent in UKB using BOLT-LMM and utilized the non-redundant SNPs ($R^2 < 0.01$) as the candidate IVs (n = 241) (S9 Table). Since UKB has a larger sample size than METSIM, the IVs identified in UKB do not have genome-wide significant associations (p-value $<5.0\times10^{-8}$) with body fat percent in METSIM. We then utilized MR-PRESSO to identify potential pleiotropy and test for the causal effect between body fat percent and the Matsuda index in the direction obesity -> insulin resistance. Noteworthy, there is no theoretical "perfect" MR method that guarantees a pleiotropy free MR analysis. Thus, the absence of evidence for pleiotropy does not necessarily prove the absence of pleiotropy. S9 Table shows the summary statistics employed in the MR analyses. The opposite causal direction, insulin resistance -> obesity, could not reliably be assessed using IVs due to the lack of genome-wide significant Matsuda index SNPs in the METSIM cohort (see Results for details).

Single nucleus RNA-sequencing and clustering

Frozen subcutaneous adipose tissue was minced over dry ice and transferred into ice cold lysis buffer consisting of 0.1% NP-40, 10mM Tris-Hcl, 10 mM NaCl, and 3 mM MgCl2. After a 10-minute incubation period, the lysate was gently homogenized using a dounce and filtered through a 70 μ m MACS smart strainer (Miltenyi Biotec #130-098-462) to remove debris. Nuclei were centrifuged at 500 g for 5 minutes at 4°C and re-suspended in wash buffer consisting of 1X PBS, 1.0% BSA, and 0.2 U/ μ l RNase inhibitor. We further filtered nuclei using a 40 μ m Flowmi cell strainer (Sigma Aldrich # BAH136800040) and centrifuged at 500 g for 5 minutes at 4°C. Pelleted nuclei were re-suspended in wash buffer and immediately processed with the 10X Chromium platform following the Single Cell 3' v2 protocol.

We used Cell Ranger [62] to build a pre-mRNA alignment reference based on the reference gencode 19 and estimate the UMIs in each cell. As the quality control, we excluded the cells that had <300 genes expressed and kept only the genes that were expressed in at least 3 cells. Then we used Seurat [63] to simultaneously cluster all the qualified cells from the 6 individuals. We identified 8 clusters and 697 signature genes (S10 Table) that have a higher expression in one of the clusters over the others.

Decomposition of adipose cell-type proportions from bulk RNA-seq data

We first used Cell Ranger to re-align the single nucleus reads to a mature mRNA reference (gencode 19) and then estimated the pseudo-bulk gene expression in the 6 individuals. Next, treating the candidate gene expression of the sn-RNA-seq data as the reference, we used MuSiC [64] to estimate the cell-type proportions from the bulk RNA-seq data. To validate the accuracy of our decomposition method, we performed both sn-RNA-seq and bulk RNA-seq using the subcutaneous adipose biopsies from the same 6 individuals. Then we predicted the cell-type proportions using the bulk RNA-seq data and compared the decomposition results to the cell-type proportions estimated from the sn-RNA-seq data of the same individuals. To ensure the independence of the test data, we used the leave-one-out strategy. In more detail,

when we estimated cell-type proportions of one individual, we used the sn-RNA-seq data from the other 5 individuals as the reference. Thus, all of the estimated cell-type proportions of each individual are based on an unrelated data set.

Because sn-RNA-seq captures not only mature mRNAs but also pre-mRNAs, the expression patterns of some genes are expected to be different between the nuclei and bulk RNA-seq data. For example, the MALAT1 gene (ENSG0000251562) exhibits an average TPM of 254 in the bulk adipose tissue in METSIM while its average TPM in the sn-RNA-seq data is 391,375. Accordingly, we observed that the decomposition results were biased by these different expression patterns when we used all the ~16,000 expressed genes as suggested by MuSiC (S4B Fig). To improve the accuracy of the decomposition, we calculated the difference in mean of the log-transformed gene expression across all genes from the target bulk RNA-seq samples and the 6 pseudo-bulk samples. Then we normalized the expression differences and kept the genes that have chi square statistic <= 1. After this filtering process, we kept ~4,000 genes that have similar expression in both the single nucleus and bulk RNA-seq data. When estimating the cell-type proportions in the bulk RNA-seq data, we used the sn-RNA-seq data from all of the 6 samples as the reference. Since cell-type proportions are estimated from RNA-seq data that is affected by technical factors, we also adjusted the proportion of each cell-type for RNA-seq technical factors when testing for the association between cell-type proportions and traits in the METSIM cohort.

QC for estimating MT gene expression

We used the same pipeline to estimate MT expression in all cohorts. First, we used FastQC [65] to verify the sequence quality of the RNA-seq data. Then, we performed a 2-pass alignment using STAR [66] (reference genome: gencode 19, hg19) and subsequently used feature-Counts [67] to estimate the TPM of each gene. Only uniquely mapped reads were counted for gene expression. MT gene expression was defined as the sum of TPMs of all MT encoded genes. Since gene expression estimates from RNA-seq data are affected by multiple technical factors [68], we corrected MT gene expression for 11 known technical factors (S11 Table) and 3 genotype PCs. We chose to correct for 3 genotype PCs to follow a similar pipeline as implemented in the GTEx project [56]. Since the GTEx RNA-seq samples were collected from deceased individuals, we also adjusted MT gene expression for the post-mortem sample collection time in the GTEx cohort.

To control for admixed ancestry in the MOSS cohort, we called variants from the RNA-seq data following the GATK pipeline. We used the recommended parameters of -window 35, -cluster 3, and filtering FS> 30 and QD <2. Furthermore, we used variants with MAF >5% and an average read depth > = 30. We then combined the MOSS and 1000 Genomes Project genotype data [69] and performed PCA, observing that the MOSS individuals clustered well with the individuals of Amerindian descent. Accordingly, we used these SNPs called from RNA-seq data to calculate the genotype PCs in the MOSS cohort for the correction of ancestry.

However, it is worth noting that the MT genome is small and has a simple structure when compared to the autosomal chromosomes. Thus, RNA metrics estimated from MT reads have a different pattern compared to that of autosomal reads. Since MT reads comprise a relatively large proportion of total reads (S11 Table), we discovered that the technical factors estimated from the RNA-seq data, such as intergenic read percent and exonic read percent, are heavily correlated with the MT read percent of each sample. S7A and S7B Fig shows that in the MET-SIM cohort, almost all the RNA metrics estimated by Picard Tools [70] show a significant association with the MT read percent (R = 0.86, p-value = 7.54×10^{-103}). Since the MT read percent the

true technical covariation. Correcting for these factors when estimating MT gene expression would thus remove signals from MT gene expression. To address this issue, we first excluded the MT reads from the RNA-seq data and then estimated these technical factors from the reads aligned to the nuclear genome using Picard Tools. The new unbiased technical factors showed much weaker associations with the MT read percent (S7C and S7D Fig).

Disentangling the associations between MT expression, body fat percent, insulin resistance (i.e. the Matsuda index), and tissue heterogeneity

Since individuals with T2D are insulin resistant and the antidiabetic medication may influence the outcome, we removed them from all analyses involving the Matsuda index. We built a multi-variable linear model treating the Matsuda index as the dependent variable to identify the associations between MT expression, BMI/body fat percent, estimated adipose cell-type proportions and Matsuda index in the METSIM cohort:

Matsuda ~
$$\beta_b * body \ fat \ percent + \sum \beta_{ci} * CT_i + \beta_M * MT$$
 (Model1)

MT indicates the corrected MT gene expression. Matsuda indicates the Matsuda index. CT_i is the estimated cell-type proportion in adipose tissue. The β s are the estimated parameters from the multi-variable linear models. Since the sum of the 8 estimated cell-type proportions equals to 1, the degree of freedom of the cell-type proportions is 7 instead of 8. We excluded the proportion of endothelial cells from the model due to its less accurate prediction when compared to the other cell-types.

To predict the Matsuda index using the other traits, we employed the following elastic net regularization [71] to select the predictors and predict the β for each variable in model 1:

$$\hat{\beta} = \operatorname*{argmin}_{\beta} (\|Matsuda - X\beta\| + \lambda \|\beta\| + \lambda \|\beta\|^2)$$

Elastic net regularization applies a penalty on the β s estimated from the predictors, and therefore reduces the estimated β s of unnecessary predictors to 0. Thus, the predictors with a non-zero estimated β are kept in the prediction model. We used the 'glmnet' package [72] to obtain the λ that has the minimum mean cross-validated error in the training data set, and then used the specified λ and β s to predict the Matsuda index. To evaluate the prediction accuracy in both models, we performed a 100-fold verification in the METSIM cohort. In more detail, we randomly split the individuals into 100 groups, and then for each group, we predicted its value based on the model that we trained with the other 99 groups. We also verified this model in 2 independent cohorts: FTC and GTEx. For building the final prediction model for these 2 cohorts, we used all individuals in the METSIM cohort as the training set and predicted the Matsuda index in GTEx and FTC as verification. Using the predicted Matsuda index, we performed a Pearson correlation test to check the association between the estimated and true Matsuda index. Since GTEx do not have the Matsuda index available, we compared the predicted Matsuda index values between the GTEx individuals with and without T2D as the verification.

Supporting information

S1 Fig. Removing the IVs associated with 4 potential confounders did not affect the MR results. (A) The associations between all IVs and three potential confounders, i.e. alcohol intake, physical activity, and smoking. The x axis shows the MR analysis in which the IVs were used. The y axis shows the p-values of the association between the IVs and confounders. (B) The MR results of body fat percent on Matsuda index without using the confounder-associated variants as IVs.

(C) The MR results of body fat percent on prediabetes without using the confounder-associated variants as IVs. (D) The MR results of prediabetes on body fat percent remained the same as in Fig 1D as none of the prediabetes IV SNPs were associated with the 3 tested confounders. (TIF)

S2 Fig. In METSIM and MOSS, we verified the significantly lower MT expression (pvalue < 0.05) in the obese individuals who have a high body fat percent or BMI compared to the lean individuals. (A) BMI is significantly associated with the adjusted MT expression in a Pearson correlation test. (B) Body fat percent is significantly associated with the adjusted MT expression in a Pearson correlation test. (C) In MOSS, obese Mexicans have a lower adjusted MT expression when compared to the lean Mexicans. (TIF)

S3 Fig. In GTEx, we used a t-test to show the association between the adjusted MT gene expression (inverse normal transformed) and T2D status in 4 different tissues. In (A) visceral adipose and (B) muscle, the adjusted MT expression is significantly higher in the nondiabetic individuals than in the T2D patients. In (C) liver and (D) blood, there is no evidence of differential MT gene expression between the non-diabetic individuals and the T2D patients. (TIF)

S4 Fig. Our QC process ensures that the SN-RNA-seq accurately estimated the cell-type proportions in subcutaneous adipose tissue. (A) The t-SNE plot shows no evidence of a batch effect in SN-RNA-seq clustering. The dots are colored by sample IDs. (B) When using all genes in the SN-RNA-seq data without any filtering, the estimated cell-type proportions are not concordant with the true cell-type proportions. Thus, using the selected genes (Fig 2B) performs much better than using all genes as reference in the decomposition process. (TIF)

S5 Fig. Person correlations show that BMI and body fat percent are highly correlated in the METSIM and FTC cohorts. Using either BMI or body fat percent predicted Matsuda index in METSIM cohort in a similar accuracy. (A) The association between the body fat percent and BMI in METSIM. (B) The association between the body fat percent and BMI in FTC. (C) The association between the predicted Matsuda index and true Matsuda index when using BMI to represent obesity status. (D) The association between the predicted Matsuda index and true Matsuda index when using body fat percent to represent obesity status. (TIF)

S6 Fig. Pearson correlations show the associations between the predictors (BMI, corrected MT expression) and Matsuda index in the METSIM and FTC cohorts. The predicted Matsuda index is always more strongly associated with the true Matsuda index than any of the predictors (Fig 4). (A) The correlation between raw BMI and the Matsuda index in METSIM. (B) The correlation between the corrected MT gene expression and Matsuda index in METSIM. (C) The correlation between raw BMI and the Matsuda index in METSIM. (C) The correlation between raw BMI and the Matsuda index in FTC. (D) The correlation between the corrected MT gene expression and Matsuda index in FTC. (TIF)

S7 Fig. Using the METSIM cohort as an example, we demonstrate that when estimating RNA metrics, including the MT reads, the RNA metrics are heavily biased by the MT read percent. When excluding the MT reads to estimate the RNA metrics, the correlations between the MT read percent and other RNA metrics are reduced. (A). The qq-plot shows the correlations between the MT read percent and estimated RNA metrics, including MT reads. The y axis shows the -log10(p-value) of the associations between MT read percent and the technical

factors estimated from the RNA-seq data. The x axis shows the expected -log10(p-value) if no true associations between the MT reads percent and technical factors exist. (B). The intergenic read percent is dominated by the MT read percent, including the MT reads. (C). The qq-plot shows the correlations between the MT read percent and estimated RNA metrics, excluding the MT reads. Compared to A, the associations between the MT read percent is not correlated with the MT read percent, excluding the MT reads. (TIF)

S1 Table. Characteristics of SN-RNA-seq samples.

(XLSX)

S2 Table. Proportions of the 8 cell types identified from 6 sn-RNA-seq adipose samples. (XLSX)

S3 Table. The MuSiC estimated cell-type proportions in the METSIM cohort. (XLSX)

S4 Table. The estimated adipose cell-type proportions are associated with the Matsuda index.

(XLSX)

S5 Table. A multi-linear model shows the significant associations between the Matsuda index and the other traits.

(XLSX)

S6 Table. The prediction model of Matsuda index. (XLSX)

S7 Table. Phenotypes of the human adipose RNA-seq cohorts. (XLSX)

S8 Table. Summary statistics of the IVs in the MR analysis between obesity (body fat percent) and prediabetes.

(XLSX)

S9 Table. Summary statistics of the IVs in the MR analysis between obesity (body fat percent) and Matsuda index.

(XLSX)

S10 Table. Signature genes identified in the 8 adipose cell-types. (XLSX)

S11 Table. Technical factors observed in the METSIM adipose RNA-seq data. (XLSX)

Acknowledgments

We thank the individuals who participated in the METSIM, GTEx, FTC, and UK Biobank as well as Jaakko Kaprio and Aila Rissanen for the contributions to the FTC study. We also thank the UNGC sequencing core at UCLA for performing RNA sequencing. The GTEx data used for the analyses described in this manuscript were obtained from the GTEx Portal on 08/14/2016 and dbGAP (accession number phs000424.v6.p1) on 08/11/2016. This research has been conducted using the UK Biobank Resource under application number 33934.

Author Contributions

Conceptualization: Zong Miao, Päivi Pajukanta.

- **Data curation:** Marcus Alvarez, Arthur Ko, Yash Bhagat, Sini Heinonen, Linda Liliana Muñoz-Hernandez, Miguel Herrera-Hernandez, Carlos Aguilar-Salinas, Teresa Tusie-Luna, Karen L. Mohlke, Markku Laakso, Kirsi H. Pietiläinen, Päivi Pajukanta.
- Formal analysis: Zong Miao.

Funding acquisition: Päivi Pajukanta.

Investigation: Zong Miao, Päivi Pajukanta.

- Methodology: Zong Miao, Marcus Alvarez, Elior Rahmani, Brandon Jew, Eran Halperin, Päivi Pajukanta.
- Project administration: Päivi Pajukanta.

Resources: Päivi Pajukanta.

Software: Zong Miao, Marcus Alvarez.

Supervision: Päivi Pajukanta.

Validation: Zong Miao, Päivi Pajukanta.

Visualization: Zong Miao.

Writing - original draft: Zong Miao, Päivi Pajukanta.

Writing – review & editing: Zong Miao, Marcus Alvarez, Arthur Ko, Yash Bhagat, Elior Rahmani, Brandon Jew, Sini Heinonen, Linda Liliana Muñoz-Hernandez, Miguel Herrera-Hernandez, Carlos Aguilar-Salinas, Teresa Tusie-Luna, Karen L. Mohlke, Markku Laakso, Kirsi H. Pietiläinen, Eran Halperin, Päivi Pajukanta.

References

- Poirier P, Giles TD, Bray GA, Hong Y, Stern JS, Pi-Sunyer FX, et al. Obesity and cardiovascular disease: Pathophysiology, evaluation, and effect of weight loss. Circulation. 2006; 113(6):898–918. https://doi.org/10.1161/CIRCULATIONAHA.106.171016 PMID: 16380542
- Zamora-Mendoza R, Rosas-Vargas H, Ramos-Cervantes MT, Garcia-Zuniga P, Perez-Lorenzana H, Mendoza-Lorenzo P, et al. Dysregulation of mitochondrial function and biogenesis modulators in adipose tissue of obese children. Int J Obes. 2018; 42(4):618–24. https://doi.org/10.1038/ijo.2017.274 PMID: 29158541
- Martinez KE, Tucker LA, Bailey BW, LeCheminant JD. Expanded normal weight obesity and insulin resistance in US adults of the national health and nutrition examination survey. J Diabetes Res. 2017; 2017. https://doi.org/10.1155/2017/9502643 PMID: 28812029
- 4. Chung JO, Cho DH, Chung DJ, Chung MY. Associations among Body Mass Index, Insulin Resistance, and Pancreatic β-Cell Function in Korean Patients with New-Onset Type 2 Diabetes FAU—Chung, Jin Ook FAU Cho, Dong Hyeok FAU—Chung, Dong Jin FAU—Chung, Min Young. Korean J Intern Med. 2012; 27(1):66–71. https://doi.org/10.3904/kjim.2012.27.1.66 PMID: 22403502
- Meah FA, DiMeglio LA, Greenbaum CJ, Blum JS, Sosenko JM, Pugliese A, et al. The relationship between BMI and insulin resistance and progression from single to multiple autoantibody positivity and type 1 diabetes among TrialNet Pathway to Prevention participants. Diabetologia. 2016; 59(6):1186– 95. https://doi.org/10.1007/s00125-016-3924-5 PMID: 26995649
- Cheng YH, Tsao YC, Tzeng IS, Chuang HH, Li WC, Tung TH, et al. Body mass index and waist circumference are better predictors of insulin resistance than total body fat percentage in middle-aged and elderly Taiwanese. Med (United States). 2017; 96(39):1–6. https://doi.org/10.1097/MD. 000000000008126 PMID: 28953643
- Neeland I. J., Turer A. T., Ayers C. R., Powell-Wiley T. M., Vega G. L., Farzaneh-Far R., ... De Lemos J. A. Dysfunctional adiposity and the risk of prediabetes and type 2 diabetes in obese adults. JAMA. 2012; 308(11), 1150–1159. https://doi.org/10.1001/2012.jama.11132 PMID: 22990274

- Goran M. I., Lane C., Toledo-Corral C., & Weigensberg M. J. Persistence of pre-diabetes in overweight and obese hispanic children; Association with progressive insulin resistance, Poor β-cell function, and increasing visceral fat. Diabetes. 2008; 57(11), 3007–3012. https://doi.org/10.2337/db08-0445 PMID: 18678615
- Tabák A. G., Herder C., Rathmann W., Brunner E. J., & Kivimäki M. Prediabetes: A high-risk state for diabetes development. The Lancet. 2012; 379(9833): 2279–2290. https://doi.org/10.1016/S0140-6736 (12)60283-9 PMID: 22683128
- Dandona P, Aljada A, Bandyopadhyay A. Inflammation the link between insulin resistance,. Trends Immunol. 2004; 25(1):4–7. https://doi.org/10.1016/j.it.2003.10.013
- Yang WM, Jeong HJ, Park SW, Lee W. Obesity-induced miR-15b is linked causally to the development of insulin resistance through the repression of the insulin receptor in hepatocytes. Mol Nutr Food Res. 2015; 59(11):2303–14. https://doi.org/10.1002/mnfr.201500107 PMID: 26179126
- Pedersen DJ, Guilherme A, Danai LV., Heyda L, Matevossian A, Cohen J, et al. A major role of insulin in promoting obesity-associated adipose tissue inflammation. Mol Metab. 2015; 4(7):507–18. <u>https:// doi.org/10.1016/j.molmet.2015.04.003</u> PMID: 26137438
- Adabimohazab R, Garfinkel A, Milam EC, Frosch O, Mangone A, Convit A. Does Inflammation Mediate the Association Between Obesity and Insulin Resistance? Inflammation. 2016; 39(3):994–1003. https:// doi.org/10.1007/s10753-016-0329-z PMID: 26956471
- Saltiel AR, Olefsky JM, Saltiel AR, Olefsky JM. Inflammatory mechanisms linking obesity and metabolic disease Find the latest version: Inflammatory mechanisms linking obesity and metabolic disease. J Clin Invest. 2017; 127(1):1–4. https://doi.org/10.1172/JCI92035 PMID: 28045402
- Li P, Oh DY, Bandyopadhyay G, Lagakos WS, Talukdar S, Osborn O, et al. LTB4 promotes insulin resistance in obese mice by acting on macrophages, hepatocytes and myocytes. Nat Med [Internet]. 2015; 21(3):239–47. https://doi.org/10.1038/nm.3800 PMID: 25706874
- Yang WM, Jeong HJ, Park SW, Lee W. Obesity-induced miR-15b is linked causally to the development of insulin resistance through the repression of the insulin receptor in hepatocytes. Mol Nutr Food Res. 2015; 59(11):2303–14. https://doi.org/10.1002/mnfr.201500107 PMID: 26179126
- Wensveen FM, Jelenčić V, Valentić S, Šestan M, Wensveen TT, Theurich S, et al. NK cells link obesityinduced adipose stress to inflammation and insulin resistance. Nat Immunol. 2015; 16(4):376–85. https://doi.org/10.1038/ni.3120 PMID: 25729921
- Pedersen DJ, Guilherme A, Danai LV., Heyda L, Matevossian A, Cohen J, et al. A major role of insulin in promoting obesity-associated adipose tissue inflammation. Mol Metab. 2015; 4(7):507–18. https://doi.org/10.1016/j.molmet.2015.04.003 PMID: 26137438
- Adabimohazab R, Garfinkel A, Milam EC, Frosch O, Mangone A, Convit A. Does Inflammation Mediate the Association Between Obesity and Insulin Resistance? Inflammation. 2016; 39(3):994–1003 <u>https://</u> doi.org/10.1007/s10753-016-0329-z PMID: 26956471
- Bycroft C, Freeman C, Petkova D, Band G, Elliott LT, Sharp K, et al. The UK Biobank resource with deep phenotyping and genomic data. Nature. 2018; 562(7726):203–9. <u>https://doi.org/10.1038/s41586-018-0579-z PMID: 30305743</u>
- Laakso M, Kuusisto J, Stančáková A, Kuulasmaa T, Pajukanta P, Lusis AJ, et al. The Metabolic Syndrome in Men study: a resource for studies of metabolic and cardiovascular diseases. J Lipid Res. 2017; 58(3):481–93. https://doi.org/10.1194/jlr.O072629 PMID: 28119442
- 22. Weisberg SP, Leibel RL, Anthony WF Jr, Weisberg SP, Mccann D, et al. Obesity is associated with macrophage accumulation in adipose tissue Find the latest version: Obesity is associated with. J Clin Invest. 2003; 112(12):1796–808. https://doi.org/10.1172/JCI19246 PMID: 14679176
- Van Harmelen V, Skurk T, Röhrig K, Lee YM, Halbleib M, Aprath-Husmann I, et al. Effect of BMI and age on adipose tissue cellularity and differentiation capacity in women. Int J Obes. 2003; <u>https://doi.org/ 10.1038/sj.ijo.0802314</u> PMID: <u>12861228</u>
- 24. Reilly SM, Saltiel AR. Adapting to obesity with adipose tissue inflammation. Nat Rev Endocrinol. 2017; 13(11):633–43. https://doi.org/10.1038/nrendo.2017.90 PMID: 28799554
- Majka SM, Miller HL, Helm KM, Acosta AS, Childs CR, Kong R, et al. Analysis and isolation of adipocytes by flow cytometry. Methods in Enzymology 2014; 537:281–296 p. https://doi.org/10.1016/B978-0-12-411619-1.00015-X PMID: 24480352
- 26. Ehrlund A, Acosta JR, Björk C, Hedén P, Douagi I, Arner P, et al. The cell-type specific transcriptome in human adipose tissue and influence of obesity on adipocyte progenitors. Sci Data. 2017; 4:1–11.
- Hagberg CE, Li Q, Kutschke M, Bhowmick D, Kiss E, Shabalina IG, et al. Flow Cytometry of Mouse and Human Adipocytes for the Analysis of Browning and Cellular Heterogeneity. Cell Rep. 2018; 24 (10):2746–2756.e5. https://doi.org/10.1016/j.celrep.2018.08.006 PMID: 30184507

- Wu M, Singh AK. Single-cell protein analysis. Curr Opin Biotechnol. 2012; 23(1):83–8. https://doi.org/10.1016/j.copbio.2011.11.023 PMID: 22189001
- Hu P, Zhang W, Xin H, Deng G. Single Cell Isolation and Analysis. Front Cell Dev Biol. 2016 Oct 25; 4 (October):135–82. https://doi.org/10.3389/fcell.2016.00116 PMID: 27826548
- Vink RG, Roumans NJ, Fazelzadeh P, Tareen SHK, Boekschoten M V., Van Baak MA, et al. Adipose tissue gene expression is differentially regulated with different rates of weight loss in overweight and obese humans. Int J Obes. 2017; 41(2):309–16. https://doi.org/10.1038/ijo.2016.201 PMID: 27840413
- Heinonen S, Buzkova J, Muniandy M, Kaksonen R, Ollikainen M, Ismail K, et al. Impaired mitochondrial biogenesis in adipose tissue in acquired obesity. Diabetes. 2015; 64(9):3135–45. <u>https://doi.org/10. 2337/db14-1937</u> PMID: 25972572
- Lindinger PW, Christe M, Eberle AN, Kern B, Peterli R, Peters T, et al. Important mitochondrial proteins in human omental adipose tissue show reduced expression in obesity. J Proteomics. 2015; 124:79–87. https://doi.org/10.1016/j.jprot.2015.03.037 PMID: 25865306
- Mardinoglu A, Kampf C, Asplund A, Fagerberg L, Hallström BM, Edlund K, et al. Defining the human adipose tissue proteome to reveal metabolic alterations in obesity. J Proteome Res. 2014; 13 (11):5106–19. https://doi.org/10.1021/pr500586e PMID: 25219818
- Vernochet C, Damilano F, Mourier A, Bezy O, Mori MA, Smyth G, et al. Adipose tissue mitochondrial dysfunction triggers a lipodystrophic syndrome with insulin resistance, hepatosteatosis, and cardiovascular complications. FASEB J. 2014; 28(10):4408–19. <u>https://doi.org/10.1096/fj.14-253971</u> PMID: 25005176
- Paglialunga S, Ludzki A, Root-McCaig J, Holloway GP. In adipose tissue, increased mitochondrial emission of reactive oxygen species is important for short-term high-fat diet-induced insulin resistance in mice. Diabetologia. 2015; 58(5):1071–80. https://doi.org/10.1007/s00125-015-3531-x PMID: 25754553
- Petersen KF, Petersen KF, Befroy D, Dufour S, Dipietro L, Cline GW, et al. Mitochondrial Dysfunction in the Elderly: Possible Role in Insulin. Science. 2007; 1140(2003):1140–3. <u>https://doi.org/10.1126/</u> science.1082889 PMID: 12750520
- Sanyal AJ, Campbell-Sargent C, Mirshahi F, Rizzo WB, Contos MJ, Sterling RK, et al. Nonalcoholic steatohepatitis: Association of insulin resistance and mitochondrial abnormalities. Gastroenterology. 2001; 120(5):1183–92. https://doi.org/10.1053/gast.2001.23256 PMID: 11266382
- American Diabetes Association. Diagnosis and Classification of Diabetes Mellitus. Diabetes Care 2010; Jan; 33(Supplement 1): S62–S69. https://doi.org/10.2337/dc10-S062 PMID: 20042775
- Shea J. L., King M. T. C., Yi Y., Gulliver W., & Sun G. Body fat percentage is associated with cardiometabolic dysregulation in BMI-defined normal weight subjects. Nutrition, Metabolism and Cardiovascular Diseases. 2012; 22(9): 741–747. https://doi.org/10.1016/j.numecd.2010.11.009 PMID: 21215604
- 40. Gómez-Ambrosi J., Silva C., Galofré J. C., Escalada J., Santos S., Gil M. J., ... Frühbeck G. Body adiposity and type 2 diabetes: Increased risk with a high body fat percentage even having a normal BMI. Obesity. 2011; 19(7): 1439–1444. https://doi.org/10.1038/oby.2011.36 PMID: 21394093
- Howard B. V., Ruotolo G., & Robbins D. C. Obesity and dyslipidemia. Endocrinology and Metabolism Clinics of North America. 2003; 32(4): 855–867. https://doi.org/10.1016/s0889-8529(03)00073-2 PMID: 14711065
- Ferrannini E. Insulin and blood pressure: Connected on a circumference? Hypertension. 2005; 45(3): 347–348. https://doi.org/10.1161/01.HYP.0000155464.44905.6c PMID: 15668355
- Verbanck M, Chen CY, Neale B, Do R. Detection of widespread horizontal pleiotropy in causal relationships inferred from Mendelian randomization between complex traits and diseases. Nat Genet. 2018; 50(5):693–8. https://doi.org/10.1038/s41588-018-0099-7 PMID: 29686387
- Bowden J., Fabiola Del Greco M., Minelli C., Smith G. D., Sheehan N. A., & Thompson J. R. Assessing the suitability of summary data for two-sample mendelian randomization analyses using MR-Egger regression: The role of the I 2 statistic. International Journal of Epidemiology. 2016; 45(6): 1961–1974. https://doi.org/10.1093/ije/dyw220 PMID: 27616674
- Lee MJ, Wu Y, Fried SK. Adipose tissue heterogeneity: Implication of depot differences in adipose tissue for obesity complications. Mol Aspects Med. 2013; 34(1):1–11. https://doi.org/10.1016/j.mam.2012. 10.001 PMID: 23068073
- Lynes MD, Tseng YH. Deciphering adipose tissue heterogeneity. Ann N Y Acad Sci. 2018; 1411(1):5– 20. https://doi.org/10.1111/nyas.13398 PMID: 28763833
- Kaaman M, Sparks LM, Van Harmelen V, Smith SR, Sjölin E, Dahlman I, et al. Strong association between mitochondrial DNA copy number and lipogenesis in human white adipose tissue. Diabetologia. 2007; 50(12):2526–33. https://doi.org/10.1007/s00125-007-0818-6 PMID: 17879081

- Heinonen S, Buzkova J, Muniandy M, Kaksonen R, Ollikainen M, Ismail K, et al. Impaired mitochondrial biogenesis in adipose tissue in acquired obesity. Diabetes. 2015; 64(9):3135–45. <u>https://doi.org/10.</u> 2337/db14-1937 PMID: 25972572
- Zou H, Hastie T. Regularization and variable selection via the elastic net. J R Stat Soc Ser B Stat Methodol. 2005; 67(5):768.
- Holmes M V., Lange LA, Palmer T, Lanktree MB, North KE, Almoguera B, et al. Causal effects of body mass index on cardiometabolic traits and events: A Mendelian randomization analysis. Am J Hum Genet. 2014; 94(2):198–208. https://doi.org/10.1016/j.ajhg.2013.12.014 PMID: 24462370
- Khera A V., Chaffin M, Wade KH, Zahid S, Brancale J, Xia R, et al. Polygenic Prediction of Weight and Obesity Trajectories from Birth to Adulthood. Cell. 2019; 177(3):587–596.e9. https://doi.org/10.1016/j. cell.2019.03.028 PMID: 31002795
- Stancáková A, Javorsky M, Kuulasmaa T, Haffner SM, Kuusisto J, Stančáková A, et al. Changes in Insulin Sensitivity and Insulin Release in Relation to Glycemia and Glucose Tolerance in 6,414 Finnish Men. Diabetes. 2009; 58(5):1212–21. https://doi.org/10.2337/db08-1607 PMID: 19223598
- Risso D., Ngai J., Speed T. P., & Dudoit S. Normalization of RNA-seq data using factor analysis of control genes or samples. Nature Biotechnology.2014; 32(9): 896–902. <u>https://doi.org/10.1038/nbt.2931</u> PMID: 25150836
- Matsuda M, DeFronzo R. Insulin Sensitivity Indices Obtained From Comparison with the euglycemic insulin clamp. Diabetes Care. 1999; 22(9):1462–70. <u>https://doi.org/10.2337/diacare.22.9.1462</u> PMID: 10480510
- Lonsdale J, Thomas J, Salvatore M, Phillips R, Lo E, Shad S, et al. The Genotype-Tissue Expression (GTEx) project. Nat Genet. 2013; 45(6):580–5. https://doi.org/10.1038/ng.2653 PMID: 23715323
- 56. Ardlie KG, Deluca DS, Segre a. V., Sullivan TJ, Young TR, Gelfand ET, et al. The Genotype-Tissue Expression (GTEx) pilot analysis: Multitissue gene regulation in humans. Science 2015; 348 (6235):648–60. https://doi.org/10.1126/science.1262110 PMID: 25954001
- 57. Jukarainen S, Heinonen S, Rämö JT, Rinnankoski-Tuikka R, Rappou E, Tummers M, et al. Obesity is associated with low nad+/sirt pathway expression in adipose tissue of BMI-discordant monozygotic twins. J Clin Endocrinol Metab. 2016; 101(1):275–83. https://doi.org/10.1210/jc.2015-3095 PMID: 26574954
- 58. Granér M, Seppälä-Lindroos A, Rissanen A, Hakkarainen A, Lundbom N, Kaprio J, et al. Epicardial fat, cardiac dimensions, and low-grade inflammation in young adult monozygotic twins discordant for obesity. Am J Cardiol. 2012; 109(9):1295–302. https://doi.org/10.1016/j.amjcard.2011.12.023 PMID: 22325087
- Loh P. R., Tucker G., Bulik-Sullivan B. K., Vilhjálmsson B. J., Finucane H. K., Salem R. M., Price A. L. Efficient Bayesian mixed-model analysis increases association power in large cohorts. Nature Genetics. 2015; 47(3), 284–290. https://doi.org/10.1038/ng.3190 PMID: 25642633
- Loh P. R., Kichaev G., Gazal S., Schoech A. P. & Price A. L. Mixed-model association for biobank-scale datasets. Nature Genetics. 2018; 50, 906–908. <u>https://doi.org/10.1038/s41588-018-0144-6</u> PMID: 29892013
- Bowden J., Smith G. D., & Burgess S. Mendelian randomization with invalid instruments: Effect estimation and bias detection through Egger regression. International Journal of Epidemiology. 2015; 44(2): 512–525. https://doi.org/10.1093/ije/dyv080 PMID: 26050253
- Zheng GXY, Terry JM, Belgrader P, Ryvkin P, Bent ZW, Wilson R, et al. Massively parallel digital transcriptional profiling of single cells. Nat Commun. 2017; 8:1–12. https://doi.org/10.1038/ncomms14049 PMID: 28091601
- Butler A, Hoffman P, Smibert P, Papalexi E, Satija R. Integrating single-cell transcriptomic data across different conditions, technologies, and species. Nat Biotechnol. 2018; 36(5):411–20. https://doi.org/10. 1038/nbt.4096 PMID: 29608179
- Wang X, Park J, Susztak K, Zhang NR, Li M. Bulk tissue cell type deconvolution with multi-subject single-cell expression reference. Nat Commun. 2019; 10(1). <u>https://doi.org/10.1038/s41467-018-08023-x</u> PMID: 30670690
- 65. http://www.bioinformatics.babraham.ac.uk/projects/fastqc/
- Dobin A, Davis C a., Schlesinger F, Drenkow J, Zaleski C, Jha S, et al. STAR: Ultrafast universal RNAseq aligner. Bioinformatics. 2013; 29(1):15–21. https://doi.org/10.1093/bioinformatics/bts635 PMID: 23104886
- Liao Y, Smyth GK, Shi W. FeatureCounts: An efficient general purpose program for assigning sequence reads to genomic features. Bioinformatics. 2014; 30(7):923–30. https://doi.org/10.1093/bioinformatics/ btt656 PMID: 24227677

- Conesa A, Madrigal P, Tarazona S, Gomez-Cabrero D, Cervera A, McPherson A, et al. A survey of best practices for RNA-seq data analysis. Genome Biol. 2016; 17(1):13. <u>https://doi.org/10.1186/ s13059-016-0881-8 PMID: 26813401</u>
- **69.** The 1000 Genomes Project Consortium et al. A global reference for human genetic variation. Nature. 2015; 526: 68–74. https://doi.org/10.1038/nature15393 PMID: 26432245
- 70. http://broadinstitute.github.io/picard
- 71. Zou H, Hastie T. Erratum: Regularization and variable selection via the elastic net. J R Stat Soc Ser B Stat Methodol. 2005; 67(5):768.
- 72. Friedman J, Hastie T, Tibshirani R. Regularization paths for generalized linear models. J Stat Software. 2010; 33(1):1–3.